Author's response to reviews

Title: Malaria prevalence and mosquito net coverage in Oromia and SNNPR regions of Ethiopia

Authors:

Estifanos B Shargie (estifanos_b@yahoo.com)
Teshome Gebre (global2000@ethionet.et)
Jeremiah Ngondi (jn250@cam.ac.uk)
Patricia M Graves (ce4x4@cdc.gov)
Aryc W Mosher (anm7@cdc.gov)
Paul M Emerson (paul.emerson@emory.edu)
Yeshewamebrat Ejigsemahu (yeshe2005@yahoo.com)
Tekola Endeshaw (teko1960@yahoo.ca)
Dereje Olana (dolana@usaid.gov)
Asrat WeldeMeskel (asratwm@yahoo.com)
Admas Teferra (admast@et.afro.who.int)
Zerihun Tadesse (zerihtad@yahoo.com)
Abate Tilahun (abate_tilahun@yahoo.com)
Gedeon Yohannes (tegicho6@yahoo.com)
Frank O Richards Jr (fxr1@cdc.gov)

Version: 4 Date: 17 September 2008

Author's response to reviews: see over
Dear Editorial Production Team,

Thank you very much for your e-mail of 16 September 2008 and for agreeing to publish our manuscript in BMC Public Health. As per the request of the editorial production team, we have now made formatting changes to the manuscript. Below, we have provided a point-by-point response to your comments.

Sincerely,

Estifanos B Shargie
Corresponding Author

Title - Please use sentence case in the title, both in the manuscript file and the submission system i.e. remove all unnecessary capitals (capitalise only the first word, and proper nouns, and no full stops/periods). For example you would change `The Title of My Prestigious Publication: A Scientific Study¿. To `The title of my prestigious publication: a scientific study¿.

Response: We have now formatted the title accordingly.

Author list - The list of authors in the manuscript should be written exactly as they are in the submission system.

Response: The author list has now been edited and matches with that in the submission system.

Section headings - Please use sentence case for all headings and sub-headings in the manuscript (i.e. remove all unnecessary capitals). For example Abstract not BSTRACT, Figure legends not Figure Legends.
Response: Corrected. We have now replaced all unnecessary capitals with sentence case for all headings and sub-headings in the manuscript.

Manuscript sections - Manuscript sections should include (in the following order): Abstract; Background; Methods; Results; Discussion; Conclusions; Abbreviations (if any); Competing interests; Authors’ contributions; Acknowledgements; References; Figure legends (if any); Tables (if any); Description of Additional files (if any).

Response: All relevant sections are included in the required order.

Figure cropping - It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image.

Response: We have cropped all the figures to the best possible level.

Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions as this is the final proofing stage for your manuscript, and you will not be able to make changes after acceptance.

Response: Thanks again. We have tried to proofread the manuscript and have now uploaded the revised version.